Author’s response to reviews

Title: Depressive and Anxiety Symptoms and Associated factors among Postnatal Women in Enugu-North Senatorial District, South-East Nigeria: A Cross-sectional Study

Authors:
Olaoluwa Samson Agbaje (agbajesam@yahoo.com)
Joy Anyanwu (joy.anyanwu@unn.edu.ng)
Prince Iheanachor Umove (prince.umoke@unn.edu.ng)
Tochi Iwuagwu (tochi.iwuagu@unn.edu.ng)
Cylia Iweama (cylia.iweama@unn.edu.ng)
Eyuche Ozoemena (eyuche.ozoemena@unn.edu.ng)
Ijeoma Nnaji (cdccpubhealth2010@gmail.com)

Version: 1 Date: 18 Jul 2018

Author’s response to reviews:

1. We have re-run the analyses based on the editor's suggestions as follows:
   a. We run the analyses for both depression and anxiety separately
   b. We used multinomial logistic regression analysis (MLR) for three outcomes of depression, anxiety, and anxious-depressed

2. We have indicated the interaction terms for both the depression and anxiety models in the manuscript

3. We prepared the table for the MLR using the standard format